

TENTH EDITION

Fundamentals of **Corporate Finance**



Brealey Myers Marcus

Fundamentals of Corporate Finance

Tenth EDITION

THE McGRAW-HILL/IRWIN SERIES IN FINANCE, INSURANCE, AND REAL ESTATE

Stephen A. Ross, Franco Modigliani Professor of Financial Economics Sloan School of Management, Massachusetts Institute of Technology *Consulting Editor*

Financial Management

Block, Hirt, and Danielsen Foundations of Financial Management Sixteenth Edition

Brealey, Myers, and Allen Principles of Corporate Finance Thirteenth Edition

Brealey, Myers, and Allen Principles of Corporate Finance, Concise Second Edition

Brealey, Myers, and Marcus Fundamentals of Corporate Finance Tenth Edition

Brooks FinGame Online 5.0

Bruner Case Studies in Finance: Managing for Corporate Value Creation Eighth Edition

Cornett, Adair, and Nofsinger Finance: Applications and Theory Fourth Edition

Cornett, Adair, and Nofsinger M: Finance Third Edition

DeMello Cases in Finance Third Edition

Grinblatt (editor) Stephen A. Ross, Mentor: Influence through Generations

Grinblatt and Titman Financial Markets and Corporate Strategy Second Edition

Higgins Analysis for Financial Management Eleventh Edition

Ross, Westerfield, Jaffe, and Jordan Corporate Finance Twelfth Edition

Ross, Westerfield, Jaffe, and Jordan Corporate Finance: Core Principles and Applications Fifth Edition Ross, Westerfield, and Jordan Essentials of Corporate Finance Tenth Edition

Ross, Westerfield, and Jordan Fundamentals of Corporate Finance Eleventh Edition

Shefrin Behavioral Corporate Finance: Decisions That Create Value Second Edition

Investments

Bodie, Kane, and Marcus Essentials of Investments Tenth Edition

Bodie, Kane, and Marcus Investments Eleventh Edition

Hirt and Block Fundamentals of Investment Management Tenth Edition

Hirschey and Nofsinger Investments: Analysis and Behavior Second Edition

Jordan, Miller, and Dolvin Fundamentals of Investments: Valuation and Management Eighth Edition

Stewart, Piros, and Heisler Running Money: Professional Portfolio Management First Edition

Sundaram and Das Derivatives: Principles and Practice Second Edition

Financial Institutions and Markets

Rose and Hudgins Bank Management and Financial Services Ninth Edition

Rose and Marquis Financial Institutions and Markets Eleventh Edition Saunders and Cornett Financial Institutions Management: A Risk Management Approach Ninth Edition

Saunders and Cornett Financial Markets and Institutions Sixth Edition

International Finance

Eun and Resnick International Financial Management Eighth Edition

Real Estate

Brueggeman and Fisher Real Estate Finance and Investments Fifteenth Edition

Ling and Archer Real Estate Principles: A Value Approach Fifth Edition

Financial Planning and Insurance

Allen, Melone, Rosenbloom, and Mahoney Retirement Plans: 401(k)s, IRAs, and Other Deferred Compensation Approaches Eleventh Edition

Altfest Personal Financial Planning Second Edition

Harrington and Niehaus Risk Management and Insurance Second Edition

Kapoor, Dlabay, and Hughes Focus on Personal Finance: An Active Approach to Help You Develop Successful Financial Skills Fifth Edition

Kapoor, Dlabay, and Hughes Personal Finance Twelfth Edition

Walker and Walker Personal Finance: Building Your Future Second Edition



Fundamentals of Corporate Finance

Tenth EDITION

Richard A. Brealey London Business School

Stewart C. Myers Sloan School of Management, Massachusetts Institute of Technology

Alan J. Marcus Carroll School of Management, Boston College







FUNDAMENTALS OF CORPORATE FINANCE, TENTH EDITION

Published by McGraw-Hill Education, 2 Penn Plaza, New York, NY 10121. Copyright ©2020 by McGraw-Hill Education. All rights reserved. Printed in the United States of America. Previous editions ©2018, 2015, and 2012. No part of this publication may be reproduced or distributed in any form or by any means, or stored in a database or retrieval system, without the prior written consent of McGraw-Hill Education, including, but not limited to, in any network or other electronic storage or transmission, or broadcast for distance learning.

Some ancillaries, including electronic and print components, may not be available to customers outside the United States.

This book is printed on acid-free paper.

1 2 3 4 5 6 7 8 9 LWI 21 20 19

ISBN 978-1-260-01396-2 (bound edition) MHID 1-260-01396-0 (bound edition) ISBN 978-1-260-70390-0 (loose-leaf edition) MHID 1-260-70390-8 (loose-leaf edition)

Portfolio Manager: Charles Synovec Product Developer: Allison McCabe-Carroll Marketing Manager: Trina Maurer Content Project Managers: Fran Simon/Jamie Koch Buyer: Susan K. Culbertson Design: Matt Diamond Content Licensing Specialist: Jacob Sullivan Cover Image: ©donskarpo/Shutterstock Compositor: MPS Limited

All credits appearing on page or at the end of the book are considered to be an extension of the copyright page.

Library of Congress Cataloging-in-Publication Data

Brealey, Richard A., author. | Myers, Stewart C., author. | Marcus, Alan J., author. Fundamentals for corporate finance / Richard A. Brealey, London
Business School, Stewart C. Myers, Sloan School of Management,
Massachusetts Institute of Technology, Alan J. Marcus, Carroll School of
Management, Boston College.
Tenth Edition. | Dubuque, IA : McGraw-Hill Education, [2019] |
Revised edition of Fundamentals of corporate finance, [2018] | Includes
bibliographical references and index.
LCCN 2018047649 | ISBN 9781260013962 (student edition : alk. paper)
LCSH: Corporations—Finance.
LCC HG4026 .B6668 2019 | DDC 658.15—dc23
LC record available at https://lccn.loc.gov/2018047649

The Internet addresses listed in the text were accurate at the time of publication. The inclusion of a website does not indicate an endorsement by the authors or McGraw-Hill Education, and McGraw-Hill Education does not guarantee the accuracy of the information presented at these sites.





Dedication To Our Families





Courtesy of Richard A. Brealey



Courtesy of Stewart C. Myers

Richard A. Brealey

Professor of Finance at the London Business School

Professor Brealey is the former president of the European Finance Association and a former director of the American Finance Association. He is a fellow of the British Academy and has served as Special Adviser to the Governor of the Bank of England and as director of a number of financial institutions. Professor Brealey is also the author (with Professor Myers and Franklin Allen) of this book's sister text, *Principles of Corporate Finance* (McGraw-Hill Education).

Stewart C. Myers

Gordon Y Billard Professor of Finance at MIT's Sloan School of Management Dr. Myers is past president of the American Finance Association and a research associate of the National Bureau of Economic Research. His research has focused on financing decisions, valuation methods, the cost of capital, and financial aspects of government regulation of business. Dr. Myers is a director of The Brattle Group, Inc. and is active as a financial consultant. He is also the author (with Professor Brealey and Franklin Allen) of this book's sister text, *Principles of Corporate Finance* (McGraw-Hill Education).



Courtesy of Alan J. Marcus

Alan J. Marcus

Mario Gabelli Professor of Finance in the Carroll School of Management at Boston College

Professor Marcus's main research interests are in derivatives and securities markets. He is co-author (with Zvi Bodie and Alex Kane) of the texts *Investments* and *Essentials of Investments* (McGraw-Hill Education). Professor Marcus has served as a research fellow at the National Bureau of Economic Research. Professor Marcus also spent two years at Freddie Mac, where he helped to develop mortgage pricing and credit risk models. He currently serves on the Research Foundation Advisory Board of the CFA Institute.



This book is an introduction to corporate finance. It focuses on how companies invest in real assets, how they raise the money to pay for the investments, and how those assets ultimately affect the value of the firm. It also provides a broad overview of the financial landscape, discussing, for example, the major players in financial markets, the role of financial institutions in the economy, and how securities are traded and valued by investors. The book offers a framework for systematically thinking about most of the important financial problems that both firms and individuals are likely to confront.

Financial management is important, interesting, and challenging. It is *important* because today's capital investment decisions may determine the businesses that the firm is in 10, 20, or more years ahead. Needless to say, a firm's success or failure also depends, in large part, on its ability to find the capital that it requires.

Finance is *interesting* for several reasons. Financial decisions often involve huge sums of money. Large investment projects or acquisitions may involve billions of dollars. Also, the financial community is international and fast-moving, with colorful heroes and a sprinkling of unpleasant villains.

Finance is *challenging*. Financial decisions are rarely cut and dried, and the financial markets in which companies operate are changing rapidly. Good managers can cope with routine problems, but only the best managers can respond to change. To handle new problems, you need more than rules of thumb; you need to understand why companies and financial markets behave as they do and when common practice may not be best practice. Once you have a consistent framework for making financial decisions, complex problems become more manageable.

This book provides that framework. It is not an encyclopedia of finance. It focuses instead on setting out the basic *principles* of financial management and applying them to the main decisions faced by the financial manager. It explains how managers can make choices between investments that may pay off at different points of time or have different degrees of risk. It also describes the main features of financial markets and discusses why companies may prefer a particular source of finance.

We organize the book around the key concepts of modern finance. These concepts, properly explained, simplify the subject. They are also practical. The tools of financial management are easier to grasp and use effectively when presented in a consistent conceptual framework. This text provides that framework.

Modern financial management is not "rocket science." It is a set of ideas that can be made clear by words, graphs, and numerical examples. The ideas provide the "why" behind the tools that good financial managers use to make investment and financing decisions.

We wrote this book to make financial management clear, useful, and fun for the beginning student. We set out to show that modern finance and good financial practice go together, even for the financial novice.

Fundamentals and Principles of Corporate Finance

This book is derived in part from its sister text *Principles of Corporate Finance*. The spirit of the two books is similar. Both apply modern finance to give students a working ability to make financial decisions. However, there are also substantial differences between the two books.



First, we provide in *Fundamentals* much more detailed discussion of the principles and mechanics of the time value of money. This material underlies almost all of this text, and we spend a lengthy chapter providing extensive practice with this key concept.

Second, we use numerical examples in this text to a greater degree than in *Principles*. Each chapter presents several detailed numerical examples to help the reader become familiar and comfortable with the material.

Third, we have streamlined the treatment of most topics. Whereas *Principles* has 34 chapters, *Fundamentals* has only 25. The relative brevity of *Fundamentals* necessitates a broader-brush coverage of some topics, but we feel that this is an advantage for a beginning audience.

Fourth, we assume little in the way of background knowledge. While most users will have had an introductory accounting course, we review the concepts of accounting that are important to the financial manager in Chapter 3.

Principles is known for its relaxed and informal writing style, and we continue this tradition in *Fundamentals*. In addition, we use as little mathematical notation as possible. Even when we present an equation, we usually write it in words rather than symbols. This approach has two advantages. It is less intimidating, and it focuses attention on the underlying concept rather than the formula.

Organizational Design

Fundamentals is organized in eight parts.

Part 1 (Introduction) provides essential background material. In the first chapter, we discuss how businesses are organized, the role of the financial manager, and the financial markets in which the manager operates. We explain how shareholders with many disparate goals might all agree that they want managers to take actions that increase the value of their investment, and we introduce the concept of the opportunity cost of capital and the trade-off that the firm needs to make when assessing investment proposals. We also describe some of the mechanisms that help to align the interests of managers and shareholders. Of course, the task of increasing shareholder value does not justify corrupt and unscrupulous behavior. We, therefore, discuss some of the ethical issues that confront managers.

Chapter 2 surveys and sets out the functions of financial markets and institutions. This chapter also reviews the crisis of 2007–2009. The events of those years illustrate clearly why and how financial markets and institutions matter.

A large corporation is a team effort, so the firm produces financial statements to help the players monitor its progress. Chapter 3 provides a brief overview of these financial statements and introduces two key distinctions—between market and book values and between cash flows and profits. This chapter also discusses some of the shortcomings in accounting practice. The chapter concludes with a summary of federal taxes.

Chapter 4 provides an overview of financial statement analysis. In contrast to most introductions to this topic, our discussion is motivated by considerations of valuation and the insight that financial ratios can provide about how management has added to the firm's value.

Part 2 (Value) is concerned with valuation. In Chapter 5, we introduce the concept of the time value of money, and because most readers will be more familiar with their own financial affairs than with the big leagues of finance, we motivate our discussion by looking first at some personal financial decisions. We show how to value long-lived





streams of cash flows and work through the valuation of perpetuities and annuities. Chapter 5 also contains a short concluding section on inflation and the distinction between real and nominal returns.

Chapters 6 and 7 introduce the basic features of bonds and stocks and give students a chance to apply the ideas of Chapter 5 to the valuation of these securities. We show how to find the value of a bond given its yield, and we show how prices of bonds fluctuate as interest rates change. We look at what determines stock prices and how stock valuation formulas can be used to infer the return that investors expect. Finally, we see how investment opportunities are reflected in the stock price and why analysts focus on the price-earnings multiple. Chapter 7 also introduces the concept of market efficiency. This concept is crucial to interpreting a stock's valuation; it also provides a framework for the later treatment of the issues that arise when firms issue securities or make decisions concerning dividends or capital structure.

The remaining chapters of Part 2 are concerned with the company's investment decision. In Chapter 8, we introduce the concept of net present value and show how to calculate the NPV of a simple investment project. We then consider more complex investment proposals, including choices between alternative projects, machine replacement decisions, and decisions of when to invest. We also look at other measures of an investment's attractiveness—its internal rate of return, profitability index, and payback period. We show how the profitability index can be used to choose between investment projects when capital is scarce. The appendix to Chapter 8 shows how to sidestep some of the pitfalls of the IRR rule.

The first step in any NPV calculation is to decide what to discount. Therefore, in Chapter 9, we work through a realistic example of a capital budgeting analysis, showing how the manager needs to recognize the investment in working capital and how taxes and depreciation affect cash flows.

We start Chapter 10 by looking at how companies organize the investment process and ensure everyone works toward a common goal. We then go on to look at various techniques to help managers identify the key assumptions in their estimates, such as sensitivity analysis, scenario analysis, and break-even analysis. We explain the distinction between accounting break-even and NPV break-even. We conclude the chapter by describing how managers try to build future flexibility into projects so that they can capitalize on good luck and mitigate the consequences of bad luck.

Part 3 (Risk) is concerned with the cost of capital. Chapter 11 starts with a historical survey of returns on bonds and stocks and goes on to distinguish between the specific risk and market risk of individual stocks. Chapter 12 shows how to measure market risk and discusses the relationship between risk and expected return. Chapter 13 introduces the weighted-average cost of capital and provides a practical illustration of how to estimate it.

Part 4 (Financing) begins our discussion of the financing decision. Chapter 14 provides an overview of the securities that firms issue and their relative importance as sources of finance. In Chapter 15, we look at how firms issue securities, and we follow a firm from its first need for venture capital, through its initial public offering, to its continuing need to raise debt or equity.

Part 5 (Debt and Payout Policy) focuses on the two classic long-term financing decisions. In Chapter 16, we ask how much the firm should borrow, and we summarize bankruptcy procedures that occur when firms can't pay their debts. In



Chapter 17, we study how firms should set dividend and payout policy. In each case, we start with Modigliani and Miller's (MM's) observation that in well-functioning markets, the decision should not matter, but we use this initial observation to help the reader understand why financial managers in practice do pay attention to these decisions.

Part 6 (Financial Analysis and Planning) starts with long-term financial planning in Chapter 18, where we look at how the financial manager considers the combined effects of investment and financing decisions on the firm as a whole. We also show how measures of internal and sustainable growth help managers check that the firm's planned growth is consistent with its financing plans. Chapter 19 is an introduction to short-term financial planning. It shows how managers ensure that the firm will have enough cash to pay its bills over the coming year. Chapter 20 addresses working capital management. It describes the basic steps of credit management, the principles of inventory management, and how firms handle payments efficiently and put cash to work as quickly as possible. It also describes how firms invest temporary surpluses of cash and how they can borrow to offset any temporary deficiency. Chapter 20 is conceptually straightforward, but it contains a large dollop of institutional material.

Part 7 (Special Topics) covers several important but somewhat more advanced topics—mergers (Chapter 21), international financial management (Chapter 22), options (Chapter 23), and risk management (Chapter 24). Some of these topics are touched on in earlier chapters. For example, we introduce the idea of options in Chapter 10, when we show how companies build flexibility into capital projects. However, Chapter 23 generalizes this material, explains at an elementary level how options are valued, and provides some examples of why the financial manager needs to be concerned about options. International finance is also not confined to Chapter 22. As one might expect from a book that is written by an international group of authors, examples from different countries and financial systems are scattered throughout the book. However, Chapter 22 tackles the specific problems that arise when a corporation is confronted by different currencies.

Part 8 (Conclusion) contains a concluding chapter (Chapter 25), in which we review the most important ideas covered in the text. We also introduce some interesting questions that either were unanswered in the text or are still puzzles to the finance profession. Thus, the last chapter is an introduction to future finance courses as well as a conclusion to this one.

Routes through the Book

There are about as many effective ways to organize a course in corporate finance as there are teachers. For this reason, we have ensured that the text is modular so that topics can be introduced in different sequences.

We like to discuss the principles of valuation before plunging into financial planning. Nevertheless, we recognize that many instructors will prefer to move directly from Chapter 4 (Measuring Corporate Performance) to Chapter 18 (Long-Term Financial Planning) in order to provide a gentler transition from the typical prerequisite accounting course. We have made sure that Part 6 (Financial Analysis and Planning) can easily follow Part 1.

Similarly, we like to discuss working capital only after the student is familiar with the basic principles of valuation and financing, but we recognize that here also





many instructors prefer to reverse our order. There should be no difficulty in taking Chapter 20 out of order.

When we discuss project valuation in Part 2, we stress that the opportunity cost of capital depends on project risk. But we do not discuss how to measure risk or how return and risk are linked until Part 3. This ordering can easily be modified. For example, the chapters on risk and return can be introduced before, after, or midway through the material on project valuation.

Changes in the Tenth Edition

Users of previous editions of this book will not find dramatic changes in either the material or the ordering of topics. But, throughout, we have made the book more up to date and easier to read. Here are some of the ways that we have done this.

Beyond the Page The Beyond the Page digital extensions and applications provide additional examples, anecdotes, spreadsheet programs, and more thoroughgoing explanations of some topics. This material is very easily accessed on the web. In this edition, we have updated them and added a number of additional applications and made them easier to access. For example, the applications are seamlessly available with a click on the e-version of the book, but they are also readily accessible in the traditional hard copy of the text using the shortcut URLs provided in the margins of relevant pages.

Improving the Flow A major part of our effort in revising this text was spent on improving the flow. Often this has meant a word change here or a redrawn diagram there, but sometimes we have made more substantial changes. One example is the discussion of discounted cash flow analysis in Chapter 9. Rather than presenting a series of disconnected examples, we now illustrate the many aspects of cash flow analysis in one integrated application. The material is substantially unchanged, but we think that the flow is much improved.

Updating Major updates in this edition revolved around the implications of recent tax reform legislation. The Tax Cuts and Jobs Act of 2017 mandated substantial changes in corporate and personal tax rates as well as in the tax treatment of depreciation and investment income. All of these changes potentially affect firms' capital budgeting and financing decisions.

Of course, in each new edition we also try to ensure that any statistics are as up to date as possible. For example, since the previous edition, we have available an extra 2 years of data on security returns. These show up in the figures in Chapter 11 of the long-run returns on stocks, bonds, and bills. Measures of EVA, data on security ownership, dividend payments, and stock repurchases are just a few of the other cases where data have been brought up to date.

New Illustrative Boxes The text contains a number of boxes with illustrative real-world examples. Many of these are new. Look, for example, at the box in Chapter 2 that describes prediction markets and what they had to say about the 2016 presidential election. Or look at the box in Chapter 15 that shows how the JOBS Act of 2012 cleared the way for companies to use crowd-funding to raise up to \$50 million from small investors who wish to invest in start-up firms.





More Worked Examples We have added more worked examples in the text, many of them taken from real companies.

Specific Chapter Changes in the Tenth Edition

Here are a few of the additions to chapter material:

- **Chapter 1** contains updated and timely examples of real capital expenditure decisions by major corporations.
- Chapter 2 includes a discussion of prediction markets in the most recent presidential election.
- Chapter 3 includes updated discussions of recent changes in tax law.

- **Chapter 6** includes a new Finance in Practice box to show how to find bond information on the web.
- **Chapter 7** provides new evidence on efficient markets and some of the anomalies literature.
- **Chapter 9** now illustrates cash flow analysis in one integrated, extended example. It also discusses and provides several examples of the impact of accelerated depreciation and immediate expensing on the value of a capital investment.
- **Chapter 14** now includes more coverage of alternative sources of cash as well as extended treatment of the variety of corporate debt.
- Chapter 16 reconsiders the present value of interest tax shields at the new, lower, corporate tax rate.
- **Chapter 20** introduces the components of working capital and the determinants of the cash cycle. It then looks briefly at each of the components including short-term debt. It provides updated discussions on recent trends in the United States concerning investments in working capital.
- **Chapter 21** features numerous updates to our coverage of the market for corporate control, for example, GE's divestment of major sectors of the firm, recent activist investor initiatives, and tax inversion strategies in the wake of recent changes to tax law.

Assurance of Learning

Assurance of learning is an important element of many accreditation standards. *Fundamentals of Corporate Finance*, Tenth Edition, is designed specifically to support your assurance-of-learning initiatives. Each chapter in the book begins with a list of numbered learning objectives, which are referred to in the end-of-chapter problems and exercises. Every test bank question is also linked to one of these objectives, in addition to level of difficulty, topic area, Bloom's Taxonomy level, and AACSB skill area. Connect, McGraw-Hill's online homework solution, and *EZ Test*, McGraw-Hill's easy-to-use test bank software, can search the test bank by these and other categories, providing an engine for targeted assurance-of-learning analysis and assessment.

AACSB Statement

McGraw-Hill Education is a proud corporate member of AACSB International. Understanding the importance and value of AACSB accreditation, *Fundamentals of Corporate Finance*, Tenth Edition, has sought to recognize the curricula guidelines detailed in the AACSB standards for business accreditation by connecting selected questions in the test bank to the general knowledge and skill guidelines found in the AACSB standards.





The statements contained in *Fundamentals of Corporate Finance*, Tenth Edition, are provided only as a guide for the users of this text. The AACSB leaves content coverage and assessment within the purview of individual schools, the mission of the school, and the faculty. While *Fundamentals of Corporate Finance*, Tenth Edition, and the teaching package make no claim of any specific AACSB qualification or evaluation, we have, within the test bank, labeled selected questions according to the six general knowledge and skills areas.



Unique Features



What makes *Fundamentals of Corporate Finance* such a powerful learning tool?

Integrated Examples

Numbered and titled examples are integrated in each chapter. Students can learn how to solve specific problems step-by-step as well as gain insight into general principles by seeing how to approach and analyze different problems.



Winning Big at the Lottery

In August 2017, a Massachusetts woman bought a Powerball lottery ticket and won a record \$758.7 million. We suspect that she received unsolicited congratulations, good wishes, and requests for money from dozens of more or less worthy charities, relations, and newly devoted friends. In response, she could fairly point out that the prize wasn't really worth \$758.7 million. That sum was to be paid in 30 annual installments of about \$23 million each. Assuming that the first payment occurred at the end of 1 year, what was the present value of the prize? The interest rate at the time was about 2.7%.

The present value of these payments is simply the sum of the present values of each annual payment. But rather than valuing the payments separately, it is much easier to treat them as a 30-year annuity. To value this annuity, we multiply \$23 million by the 30-year annuity factor:

 $PV = 23 \times 30$ -year annuity factor $= 23 \times \left[\frac{1}{r} - \frac{1}{r(1+r)^{30}}\right]$

At an interest rate of 2.7%, the annuity factor is

$\left[\frac{1}{.027} - \frac{1}{.027(1.027)^{30}}\right] = 20.3829$

Spreadsheet Solutions Boxes

These boxes provide the student with detailed examples of how to use Excel spreadsheets when applying financial concepts. The boxes include questions that apply to the spreadsheet, and their solutions are given at the end of the applicable chapter. These spreadsheets are available for download in Connect.

Spreadsheet Solutions Bond Valuation

Excel and most other spreadsheet programs provide builtin functions to compute bond values and yields. They typically ask you to input both the date you buy the bond (called the settlement date) and the maturity date of the bond. The Excel function for bond value is:

=PRICE(settlement date, maturity date, annual coupon rate, yield to maturity, redemption value as percent of face value, number of coupon payments per year)

(If you can't remember the formula, just remember that you can go to the Formulas tab in Excel, and from the Financial tab pull down the PRICE function, which will prompt you for the necessary inputs.) For our 2.25% coupon bond, we would enter the values shown in the spreadsheet below. ternatively, we could simply enter the following function in ccel:

 $= \mbox{PRICE(DATE(2018.2,15), DATE(2021.2,15), 0.0255, 0.02391, 100, \mbox{1}\) The DATE function in Excel, which we use for both the settlement and maturity dates, uses the format DATE(peranomth, day).$

DAI typear,month.day). Notice that the coupon rate and yield to maturity are expressed as decimals, not percentages. In most cases, redemption value will be 100 (i.e., 100% of face value), and the resulting price will be expressed as a percent of face value. Occasionally, however, you may encounter bonds that pay off at a premium or discount to face value. One example would be callable bonds at a premium before maturity.

Excel Exhibits

Selected exhibits are set as Excel spreadsheets. The accompanying files are available for instructors and students in Connect.

| A | B | С | D | E | | | |
|--|--|--------------------|------------------------------|-------------------------------|--|--|--|
| Finding the pre | Finding the present value of multiple cash flows using a spreadsheet | | | | | | |
| | | | | | | | |
| Time until CF | Cash flow | Present value | Formula in Col C | Alternative formula for Col C | | | |
| 0 | 8000 | \$8,000.00 | =PV(\$B\$10, A4, 0, -B4) | =B4/(1 + \$B\$10)^A4 | | | |
| 1 | 4000 | \$3,703.70 | =PV(\$B\$10, A5, 0, -B5) | =B5/(1 + \$B\$10)^A5 | | | |
| 2 | 4000 | \$3,429.36 | =PV(\$B\$10, A6, 0, -B6) | =B6/(1 + \$B\$10)^A6 | | | |
| | | | | | | | |
| SUM | | \$15,133.06 | =SUM(C4:C6) | =SUM(C4:C6) | | | |
| | | | | | | | |
| Discount rate: | 0.08 | | | | | | |
| | | | | | | | |
| Notice that the time until each payment is found in column A. | | | | | | | |
| Once we enter | the formula fo | r present value in | cell C4, we can copy it to c | cells C5 and C6. | | | |
| The present value for other interest rates can be found by changing the entry in cell B10. | | | | | | | |

Finance in Practice Boxes

These are excerpts that appear in most chapters, often from the financial press, providing real-life illustrations of the chapter's topics, such as ethical choices in finance, disputes about stock valuation, financial planning, and credit analysis.

Finance in Practice Ethical Disputes in Fin

Short-Selling

Investors who take short positions are betting that securities will fall in price. Usually they do this by borrowing the security, selling it for cash, and then waiting in the hope that they will be able to buy it back cheaply. 'In 2007, hedge fund manager John Paulson took a huge short position in mortagae-backed securities. The bet paid off, and that year Paulson's trade made a profit of \$1 billion for his fund.' Was Paulson's trade unefaid? Some believe not only that he was profiting from the missery that resulted from the crash in mortane-backed securities but that his short trades

Was Paulson's trade unethical? Some believe not only that he was profiling from the missery that resulted from the crash in mortgage-backed securities, but that his short trades accentuated the collapse. It is certainly true that short-sellers have never been popular. For example, following the crash of 1929, one commentator compared short-selling to the ghoulishness of "creatures who, at all great earthquakes and fres, spring up to rob broken homes and injured and dead But sometimes raids can enhance shareholder value. For example, in 2012 and 2013. Relational Investors teamed up with the California State Teachers' Retirement System (CSTRS, a pension fund) to try to force Timken Co. to spill into two separate companies, one for its steel business and one for its industrial bearings business. Relational and CSTRS believed that Timken's combination of unrelated businesses was unfocused and inefficient. Timken management responded that breakup would "deprive our shareholders of long-run valueall in an attempt to create illusory short-term gains through financial engineering". But Timken's stock price rose at the prospect of a breakup, and a nonbinding shareholder vote on Relational's proposal attracted a 53% majority. Finally, in 2014 Timken spun off its steel business in a new corporation, Timken Steel

How do you draw the ethical line in such examples? Was

Financial Calculator Boxes and Exercises

In a continued effort to help students grasp the critical concept of the time value of money, many pedagogical tools have been added throughout the first section of the text. Financial Calculator boxes provide examples for solving a variety of problems, with directions for the most popular financial calculators.

Financial Calculator Using a Financial Calculator to Compute Bond Yield

You can use a financial calculator to calculate the yield to maturity on our 2.25\% Treasury bond. The inputs are:

| | n | i | PV | РМТ | FV |
|---------|---|-------|---------|------|------|
| Inputs | 3 | | 995.938 | 22.5 | 1000 |
| Compute | | 2.392 | | | |

Now compute *i* and you should get an answer of 2.392%, which is just a tick above the yield reported in Table 6.1. Let's now redo this calculation but recognize that the coupons are paid semiannually. Instead of three annual coupon payments of \$22.50, the bond makes six semiannual

payments of \$11.25. Therefore, we can find the semiannual yield as follows:

| | n | i | PV | РМТ | FV |
|---------|---|---------|---------|-------|------|
| Inputs | 6 | | 995.938 | 11.25 | 1000 |
| Compute | | 1.19556 | | | |

This yield to maturity, of course, is a 6-month yield, not an annual one. Bond dealers would typically annualize the semiannual rate by doubling it, so the yield to maturity would be quoted as 119556 \times 2 = 2.391%, which exactly matches the value in Table 61.

Self-Test Questions

Provided in each chapter, these helpful questions enable students to check their understanding as they read. Answers are worked out at the end of each chapter.

6.4 Self-Test

Suppose that the market interest rate is 4% and then drops overnight to 2%. Calculate the present values of the 2.25%, 3-year bond and of the 2.25%, 30-year bond both before and after this change in interest rates. Assume annual coupon payments. Confirm that your answers correspond with Figure 6.5. Use your financial calculator or a spreadsheet. You can find a box on bond pricing using Excel later in this chapter.

"Beyond the Page" Interactive **Content and Applications**

Additional resources and hands-on applications are just a click away. Students can tap or click the icons in the e-version or use the direct web links to learn more about key concepts and try out calculations, tables, and figures when they go "Beyond the Page."



A change in interest rates has only a modest impact on the present value of near-term cash flows but a much greater impact on the value of distant cash flows. Therefore, any change has a greater impact on the price of long-term bonds than the price of shortterm bonds. For example, compare the two curves in Figure 6.5. The blue line shows how the value of the 3-year, 2.25% coupon bond varies with the interest rate. The green line shows how the price of a 30-year, 2.25% bond varies. You can see that the 30-year bond is more sensitive to interest rate fluctuations than the 3-year bond. This should not surprise you. If you buy a 3-year bond and rates then rise, you will be stuck with a bad deal-vou could have got a better interest rate if you had waited. However, think how much worse it would be if the loan had been for 30 years rather than 3 years. The longer the loan, the more you have lost by accepting what turns out to be a low interest rate. This shows up in a bigger decline in the price of the longer-term bond. Of course, there is a flip side to this effect, which you can also see from Figure 6.5. When interest rates fall, the longer-term bond responds with a greater increase in price.

Web Exercises

Select chapters include Web Exercises that allow students to utilize the Internet to apply their knowledge and skills with real-world companies.

WEB EXERCISES

- Log on to www.investopedia.com to find a simple calculator for working out bond prices. (Start by clicking the *Investing* link and then look for another link to *Calculators*.) Check 1. Log on to www whether a change in yield has a greater effect on the price of a long-term or a short-term bond.
- 2. When we plotted the yield curve in Figure 6.7, we used the prices of Treasury strips. You can find current prices of strips by logging on to The Wall Street Journal website (www.wsj.com) and clicking on Markets. Market Data, and then Rates. Try plotting the yields on stripped coupons against maturity. Do they currently increase or decline with maturity? Can you explain why? You can also use The Wall Street Journal site to compare the yields on nominal Treasury by the data or STIP for the Wall Street Journal site to compare the yields on nominal Treasury. bonds with those on TIPS. Suppose that you are confident that inflation will be 3% per year. Which bonds are the better buy
- 3. In Figure 6.9, we showed how bonds with greater credit risk have promised higher yields to maturity. This yield spread goes up when the economic outlook is particularly uncertain. You can check how much extra yield lower-grade bonds offer today by logging on to the Federal Reserve Economic Database (FRED) at the St. Louis Fed website (fred.stouisfed.org). Search for Corporate Book and composite yields on the product and Baa bond. How does the current spread out the statement of the statem and Baa bond How does the current spread

Minicases

Integrated minicases allow students to apply their knowledge to relatively complex, practical problems and typical real-world scenarios.

MINICASE

savings account intact for unexpected expenses or emergencies

Old Alfred Road, who is well-known to drivers on the Maine Turnpike, has reached his 70th birthday and is ready to retire. Mr. Road has no formal training in finance but has saved his moust invested careffully.
Mr. Road owns his home—the mortgage is paid off—and does not want to moves. He is a widower, and he wants to bequeat hit house and any remaining assets to his daughter.
He has accumulated savings of \$180,000, conservatively investend: are investment are yielding 9% interest. Mr. Road als has \$12,000 in a savings account at 5% interest. He wants to keep varings account nata for unexpected expenses or emergencies.
Suppose Mr. Road will live for 20 more years and is willing to

Suppose Mr. Road will live for 20 more years and is willing to

Assume that the investment portfolio continues to

Supplements

In addition to the overall refinement and improvement of the text material, considerable effort was put into developing an exceptional supplement package to provide students and instructors with an abundance of teaching and learning resources.

Instructor Library

The Connect Instructor Library is your repository for additional resources to improve student engagement in and out of class. You can select and use any asset that enhances your lecture. The Connect Instructor Library includes all of the instructor supplements for this text.

Solutions Manual

Andrew Hession-Kunz worked with the authors to prepare this resource containing detailed and thoughtful solutions to all the end-of-chapter problems.

Instructor's Manual

This manual, updated and enhanced by Matthew Will at the University of Indianapolis, includes a descriptive preface containing alternative course formats and case teaching methods, a chapter overview and outline, key terms and concepts, a description of the PowerPoint slides, video teaching notes, related web links, and pedagogical ideas.

Test Bank

Matthew Will has also thoroughly reviewed and revised the test bank, adding new questions and ensuring that all of the content is closely correlated to the text. More than 2,000 true/false, multiplechoice, and discussion questions/problems are available to the instructor at varying levels of difficulty and comprehension. All questions are tagged by learning objective, topic, AACSB category, and Bloom's Taxonomy level. Complete answers are provided for all test questions and problems. The test bank is available as downloadable Word files, and tests can also be created online within McGraw-Hill's Connect or through TestGen.

TestGen is a complete, state-of-theart test generator and editing application software that allows instructors to quickly and easily select test items from McGraw-Hill's test bank content. The instructors can then organize, edit, and customize questions and answers to rapidly generate tests for paper or online administration. Questions can include stylized text, symbols, graphics, and equations that are inserted directly into questions using built-in mathematical templates. TestGen's random generator provides the option to display different text or calculated number values each time questions are used. With both quick-and-simple test creation and flexible and robust editing tools, TestGen is a complete test generator system for today's educators.

PowerPoint Presentations

These visually stimulating slides have been fully updated by Matthew Will, with colorful graphs, charts, and lists. The slides can be edited or manipulated to fit the needs of a particular course.

Beyond the Page Content

The authors have created a wealth of additional examples, explanations, and applications, available for quick access by instructors and students. Each "Beyond the Page" feature is called out in the text with an icon that links directly to the content.

Excel Solutions and Templates

Excel templates are available in Connect for select exhibits and various end-of-chapter problems that have been set as Excel spreadsheets. They correlate with specific concepts in the text and allow students to work through financial problems and gain experience using spreadsheets. Also refer to the valuable Spreadsheet Solutions Boxes that are sprinkled throughout the text for some helpful prompts on working in Excel.

Student Study Center

The Connect Student Study Center is the place for students to access additional resources. The Student Study Center

- Offers students quick access to the Beyond the Page features, Excel files and templates, lectures, eBooks, and more.
- Provides instant practice material and study questions, easily accessible on the go.

Student Progress Tracking

Connect keeps instructors informed about how each student, section, and class is performing, allowing for more productive use of lecture and office hours. The progresstracking function enables you to

- View scored work immediately and track individual or group performance with assignment and grade reports.
- Access an instant view of student or class performance relative to learning objectives.
- Collect data and generate reports required by many accreditation organizations, such as AACSB and AICPA.

McGraw-Hill Customer Care Contact Information

At McGraw-Hill, we understand that getting the most from new technology can be challenging. That's why our services don't stop after you purchase our products. You can e-mail our product specialists 24 hours a day to get product training online. Or you can search our knowledge bank of frequently asked questions on our support website.

For customer support, call **800-331-5094** or visit www.mhhe.com/support. One of our technical support analysts will be able to assist you in a timely fashion.



Students—study more efficiently, retain more and achieve better outcomes. Instructors-focus

SUCCESSFUL SEMESTERS INCLUDE CONNECT

FOR INSTRUCTORS

You're in the driver's seat.

Want to build your own course? No problem. Prefer to use our turnkey, prebuilt course? Easy. Want to make changes throughout the semester? Sure. And you'll save time with Connect's auto-grading too.



Less Time Grading

| 12.1 Evolution Acts on Populations | Page | 238 / 826 | | |
|---------------------------------------|---|-----------|--|--|
| 26 | But what is evolution? A simple definition of evolution \Box is descent with modification. "Descent" implies inheritance, "modification" refers to change in traits from generation to generation. For example, we see evolution at work in the lines, tigers, and leoperds that descended from one accentral cat species. | | | |
| 12.2 Evolutionary Thought Has | Evolution has another, more specific, definition as well. Recall from chapter 7 @ that a gase is a DNA sequence that encodes a protein; in part, an organism's proteins determine its traits. Moreover, each gree can have multiple versions, callelse. We have also seen that a population [] consists of interferenting members of the same | | | |
| Evolved for Centuries | species (see figure 1.2 (g)). Biologists say that revolution occurs in a population when some alleles become more contrator, and course less common, freem one generation to the text. A more precise definition of evolution, then, is genetic adapt in a population over multiple generations. | | | |
| 01 41 01 061 01 01 41 | According to this definition, evolution is detectable by examining a population's gree pool [3]—Instrumer collection of grees and their alleles. Evolution is a change in allele frequencies [3] and allele's frequency is evolutioned at the number of courses of their after the three consolution. | | | |
| 12.3 Naturel Selection Molds | Suppose, for example, that a gene has 2 possible alleles, A and a. In a population of 100 dipoid individuals, the gree has 200 alleles. If 160 of those alleles are a, then the frequency of a is 160/200, or 0.8. In the next generation, a may become either more or less common. Become and individual's alleles do not chance, evolution | | | |
| | Previous Highlight 🔇 Previous Section Next Section > Next Highlight 🖄 🙀 A | A 🔒 | | |

They'll thank you for it.

Adaptive study resources like SmartBook[®] help your students be better prepared in less time. You can transform your class time from dull definitions to dynamic debates. Hear from your peers about the benefits of Connect at www.mheducation.com/highered/connect

Make it simple, make it affordable.

Connect makes it easy with seamless integration using any of the major Learning Management Systems—Blackboard®, Canvas, and D2L, among others-to let you organize your course in one convenient location. Give your students access to digital materials at a discount with our inclusive access program. Ask your McGraw-Hill representative for more information.



©Hill Street Studios/Tobin Rogers/Blend Images LLC



Solutions for your challenges.

A product isn't a solution. Real solutions are affordable, reliable, and come with training and ongoing support when you need it and how you want it. Our Customer Experience Group can also help you troubleshoot tech problems—although Connect's 99% uptime means you might not need to call them. See for yourself at status.mheducation.com

FOR STUDENTS

Effective, efficient studying.

Connect helps you be more productive with your study time and get better grades using tools like SmartBook, which highlights key concepts and creates a personalized study plan. Connect sets you up for success, so you walk into class with confidence and walk out with better grades.



©Shutterstock/wavebreakmedia

⁶⁶ I really liked this app—it made it easy to study when you don't have your textbook in front of you.⁹⁹

> - Jordan Cunningham, Eastern Washington University

Study anytime, anywhere.

Download the free ReadAnywhere app and access your online eBook when it's convenient, even if you're offline. And since the app automatically syncs with your eBook in Connect, all of your notes are available every time you open it. Find out more at **www.mheducation.com/readanywhere**



No surprises.

The Connect Calendar and Reports tools keep you on track with the work you need to get done and your assignment scores. Life gets busy; Connect tools help you keep learning through it all.



Learning for everyone.

McGraw-Hill works directly with Accessibility Services Departments and faculty to meet the learning needs of all students. Please contact your Accessibility Services office and ask them to email accessibility@mheducation.com, or visit **www.mheducation.com/about/accessibility.html** for more information.

Acknowledgments

We take this opportunity to thank all of the individuals who helped us prepare this and previous editions. We want to express our appreciation to those instructors whose insightful comments and suggestions were invaluable to us during this revision.

Marlena Akhbari Wright State University

Timothy Alzheimer Montana State University

Tom Arnold University of Richmond

Robert Balik Western Michigan University

Anindam Bandopadhyaya University of Massachusetts–Boston

Chenchu Bathala Cleveland State University

Richard Bauer Saint Mary's University

LaDoris Baugh Athens State University

John R. Becker Blease Washington State University–Vancouver

Theologos Bonitsis New Jersey Institute of Technology

Stephen Borde University of Central Florida

Edward Boyer Temple University

Stephen Buell Lehigh University

Deanne Butchey Florida International University

Shelley Canterbury George Mason University

Michael Casey University of Central Arkansas Fan Chen University of Mississippi

Nicole Choi Washington State University–Pullman

Bruce Costa University of Montana

Ernie Dancer Columbus State Community College

Kenneth Daniels Virginia Commonwealth University

Morris Danielson Saint Joseph's University

Natalya Delcoure Sam Houston State University

Jared DeLisle Washington State University–Vancouver

Steven Dennis University of North Dakota

Robert Dubil University of Utah–Salt Lake City

Alan D. Eastman Indiana University of Pennsylvania

Michael Ehrlich New Jersey Institute of Technology

Richard Elliot University of Utah–Salt

Mike Evans Winthrop University

Lake City

James Falter Franklin University John Fay Santa Clara University

Richard Fedler Georgia State University

Michael Ferguson University of Cincinnati

> Dov Fobar Brooklyn College

> Eric Fricke

California State University– East Bay Steve Gallaher

Southern New Hampshire University Sharon Garrison

University of Arizona

Ashley Geisewite Southwest Tennessee Community College

Homaifar Ghassem Middle Tennessee State University

Phillip Giles Columbia University

Gary Gray Penn State University– University Park

John Halstead Liberty University

Mahfuzul Haque Indiana State University

Andrew Hession-Kunz Boston College

Larry Holland University of Arkansas–Little Rock

James J. Hopper Mississippi State University Jian "Emily" Huang California State University–Chico

Stoyu Ivanov San Jose State University

Raymond Jackson University of Massachusetts–Dartmouth

Keith Jacob University of Montana

Bharat Jain Towson University

Benjamas Jirasakuldech Slippery Rock University of Pennsylvania

Mark Johnson Loyola University Maryland

Steve Johnson Sam Houston State University

Daniel Jubinski Saint Joseph's University

Alan Jung San Francisco State University

Ayala Kayhan Louisiana State University

Marvin Keene Coastal Carolina University

Eric Kelley University of Arizona

Dong Man Kim California State University– San Bernardino

Jullavut Kittiakarasakun University of Texas at San Antonio

Ladd Kochman Kennesaw State University David Kuipers University of Missouri– Kansas City

Mark Lane Hawaii Pacific University–Honolulu

Linda Lange Regis University

Doug Letsch Walden University

Paul Lewandowski Saginaw Valley State University

Yuming Li California State University–Fullerton

Qianqiu Liu University of Hawaii–Manoa

Sheen Liu Washington State University–Vancouver

Wilson Liu James Madison University

Scott W. Lowe James Madison University

Qingzhong Ma Cornell University–Ithaca

Yulong Ma California State University– Long Beach

Brian Maris Northern Arizona University

Jinghan Meng University of North Carolina–Chapel Hill

Jose Mercardo University of Central Missouri

Paulo Miranda Purdue University–Calumet Hammond Derek Mohr State University of New York at Buffalo

Helen Moser University of Minnesota– Minneapolis

Tammie Mosley California State University– East Bay

Vivian Nazar Ferris State University

Steve Nenninger Sam Houston State University

Srinivas Nippani Texas A&M University–Commerce

Prasad Padmanabhan Saint Mary's University

Ohaness Paskelian University of Houston–Downtown

Jeffrey Phillips Colby-Sawyer College

Richard Ponarul California State University–Chico

Gary Porter John Carroll University

Eric Powers University of South Carolina

Ronald Prange Western Michigan University

Robert Puelz Southern Methodist University

Sunder Raghavan Embry-Riddle University

Vedpauri Raghavan Embry-Riddle University– Daytona Beach Ganas K. Rakes *Ohio University*

Adam Reed University of North Carolina–Chapel Hill

Thomas Rhee California State University– Long Beach

Jong Rhim University of Southern Indiana

Joe Riotto New Jersey City University

Mukunthan Santhanakrishnan Idaho State University

Maria Schutte Michigan Technological University

Adam Schwartz Washington & Lee University

John Settle Portland State University

Michael G. Sher Metropolitan State University

Henry Silverman Roosevelt University

Ron Spicer Colorado Tech University

Roberto Stein Tulane University

Tom Strickland Middle Tennessee State University

Jan Strockis Santa Clara University

Joseph Tanimura San Diego State University

Steve Tokar University of Indianapolis Damir Tokic University of Houston–Downtown

Michael Toyne Northeastern State University

James Turner Weber State University

Bonnie Van Ness University of Mississippi

Joe Walker University of Alabama–Birmingham

Kenneth Washer Texas A&M University–Commerce

K. Matthew Wong St. John's University

Qiang Wu Rensselaer Polytechnic University

David Yamoah Kean University

Fred Yeager St. Louis University

Kevin Yost Auburn University

Emilio R. Zarruk Florida Atlantic University

Athena Zhang California State University–Chico

Shaorong Zhang Marshall University

Yilei Zhang University of North Dakota

Yiling Zhang University of Texas–Arlington

Zhong-Guo Zhou California State University–Northridge



In addition, we would like to thank the dedicated experts who have helped with updates to our instructor materials and online content in Connect and LearnSmart, including Andrew Hession-Kunz, Matt Will, Marc-Anthony Isaacs, and Nicholas Racculia. Their efforts are much appreciated as they will help both students and instructors. We also appreciate help from Aleijda de Cazenove Balsan and Malcolm Taylor.

We are grateful to the talented staff at McGraw-Hill Education, especially Allison McCabe-Carroll, Senior Product Developer; Chuck Synovec, Director, Finance; Kevin Moran, Director, Digital Content; Fran Simon and Jamie Koch, Content Project Managers; Matthew Diamond, Senior Designer; Trina Mauer, Senior Marketing Manager; and Dave O'Donnell, Marketing Specialist.

Finally, as was the case with the last nine editions, we cannot overstate the thanks due to our families.

Richard A. Brealey Stewart C. Myers Alan J. Marcus

Contents in Brief

| Part One | 1 | Goals and Governance of the Corporation 2 |
|---------------------------|---------------------|--|
| Introduction | 2 | Financial Markets and Institutions 32 |
| | 3 | Accounting and Finance 56 |
| | 4 | Measuring Corporate Performance 86 |
| Part Two | 5 | The Time Value of Money 118 |
| Value | 6 | Valuing Bonds 166 |
| | 7 | Valuing Stocks 196 |
| | 8 | Net Present Value and Other Investment Criteria 238 |
| | 9 | Using Discounted Cash-Flow Analysis to Make Investment Decisions 274 |
| | 10 | Project Analysis 304 |
| Part Three | 11 | Introduction to Risk, Return, and the Opportunity Cost of Capital 332 |
| Risk | 12 | Risk, Return, and Capital Budgeting 362 |
| | 13 | The Weighted-Average Cost of Capital and Company Valuation 392 |
| Part Four | 14 | Introduction to Corporate Financing 420 |
| Financing | 15 | How Corporations Raise Venture Capital and Issue Securities 440 |
| Part Five | 16 | Debt Policy 466 |
| Debt and Payout Policy | 17 | Payout Policy 504 |
| | | |
| Part Six | 18 | Long-Term Financial Planning 528 |
| Financial Analysis | 19 | Short-Term Financial Planning 550 |
| and Planning | 20 | Working Capital Management 572 |
| Part Seven | 21 | Mergers, Acquisitions, and Corporate Control 612 |
| Special Topics | 22 | International Financial Management 640 |
| | 23 | Options 666 |
| | 24 | Risk Management 694 |
| Part Eight Conclusion | 25 | What We Do and Do Not Know about Finance 714 |
| | App Glos Inde | pendix: Present Value and Future Value Tables A-1 ssary G-1 ex IND-1 |
| | | |



This page intentionally left blank

Contents

Part One

Introduction

Chapter 1

Goals and Governance of the Corporation 2

- 1.1 Investment and Financing Decisions 4 The Investment (Capital Budgeting) Decision 6 The Financing Decision 6
- 1.2 What Is a Corporation? 8 Other Forms of Business Organization 9
- Who Is the Financial Manager? 10 13
- 1.4 Goals of the Corporation 12 Shareholders Want Managers to Maximize Market Value 12
- Agency Problems, Executive Compensation, 1.5 and Corporate Governance 15 Executive Compensation 16 Corporate Governance 17
- 1.6 The Ethics of Maximizing Value 18
- 1.7 Careers in Finance 21
- 1.8 Preview of Coming Attractions 22
- Snippets of Financial History 23 1.9

Summary 25 Questions and Problems 26

Chapter 2

2.4

Financial Markets and Institutions 32

- The Importance of Financial Markets 21 and Institutions 34 The Flow of Savings to Corporations 35 2.2 The Stock Market 37
- Other Financial Markets 38 Financial Intermediaries 40 Financial Institutions 42 Total Financing of U.S. Corporations 44

2.3 Functions of Financial Markets and Intermediaries 45 Transporting Cash across Time 45 Risk Transfer and Diversification 45 Liquidity 46 The Payment Mechanism 47 Information Provided by Financial Markets 47 The Crisis of 2007–2009 49

Summary 51

Questions and Problems 52

Chapter 3 Accounting and Finance

- The Balance Sheet 58 3.1 Book Values and Market Values 61
- 3.2 The Income Statement 63 Income versus Cash Flow 64
- 3.3 The Statement of Cash Flows 67 Free Cash Flow 69
- Accounting Practice and Malpractice 70 3.4

56

3.5 Taxes 73 Corporate Tax 73 Personal Tax 74

> Summary 75 Questions and Problems 76

Chapter 4

Measuring Corporate Performance 86

- 4.1 How Financial Ratios Relate to Shareholder Value 88
- Measuring Market Value and Market Value 42 Added 89
- 4.3 Economic Value Added and Accounting Rates of Return 91 Accounting Rates of Return 93 Problems with EVA and Accounting Rates of Return 95
- 4.4 Measuring Efficiency 96
- 4.5 Analyzing the Return on Assets: The Du Pont System 98 The Du Pont System 98
- 4.6 Measuring Financial Leverage 100 Leverage and the Return on Equity 102
- 4.7 Measuring Liquidity 103
- 4.8 Interpreting Financial Ratios 104
- The Role of Financial Ratios 108 4.9

Summary 109 Questions and Problems 110 Minicase 116

Part Two Value

Chapter 5 The Time Value of Money 118

5.1 Future Values and Compound Interest 120

Contents

- 5.2 Present Values 123 Finding the Interest Rate 127
- 5.3 Multiple Cash Flows 128 Future Value of Multiple Cash Flows 128 Present Value of Multiple Cash Flows 129
- 5.4 Reducing the Chore of the Calculations: Part 1 131 Using Financial Calculators to Solve Simple Time-Value-of-Money Problems 131 Using Spreadsheets to Solve Simple Time-Value-of-Money Problems 132
- 5.5 Level Cash Flows: Perpetuities and Annuities 135 How to Value Perpetuities 135 How to Value Annuities 136 Future Value of an Annuity 140 Annuities Due 143
- 5.6 Reducing the Chore of the Calculations: Part 2 144 Using Financial Calculators to Solve Annuity Problems 144 Using Spreadsheets to Solve Annuity Problems 145

5.7 Effective Annual Interest Rates 145

5.8 Inflation and the Time Value of Money 147 Real versus Nominal Cash Flows 147 Inflation and Interest Rates 149 Valuing Real Cash Payments 151 Real or Nominal? 152

> Summary 152 Questions and Problems 153 Minicase 164

Chapter 6 Valuing Bonds 166

- 6.1 The Bond Market 168 Bond Characteristics 168
- 6.2 Interest Rates and Bond Prices 170 How Bond Prices Vary with Interest Rates 172 Interest Rate Risk 174
- 6.3 Yield to Maturity 174 Calculating the Yield to Maturity 176
- 6.4 Bond Rates of Return 176
- 6.5 The Yield Curve 178 Nominal and Real Rates of Interest 181
- 6.6 Corporate Bonds and the Risk of Default 182 Protecting against Default Risk 185 Not All Corporate Bonds Are Plain Vanilla 187

Summary 187 Questions and Problems 188

Chapter 7 Valuing Stocks 196

- 7.1 Stocks and the Stock Market 198 Reading Stock Market Listings 199
- 7.2 Market Values, Book Values, and Liquidation Values 201
- 7.3 Valuing Common Stocks 203 Valuation by Comparables 203 Price and Intrinsic Value 204 The Dividend Discount Model 206
- 7.4 Simplifying the Dividend Discount Model 209

 Case 1: The Dividend Discount Model with No Growth 209
 Case 2: The Dividend Discount Model with Constant Growth 209
 Case 3: The Dividend Discount Model with Nonconstant Growth 214
- 7.5 Valuing a Business by Discounted Cash Flow 218 Valuing the Concatenator Business 218 Repurchases and the Dividend Discount Model 219
- 7.6 There Are No Free Lunches on Wall Street 220 Random Walks and Efficient Markets 221
- 7.7 Market Anomalies and Behavioral Finance 224 Market Anomalies 224 Bubbles and Market Efficiency 226 Behavioral Finance 227

Summary 228 Questions and Problems 229 Minicase 236

Chapter 8

Net Present Value and Other Investment Criteria 238

- 8.1 Net Present Value 240 A Comment on Risk and Present Value 241 Valuing Long-Lived Projects 242 Choosing between Alternative Projects 244
- 8.2 The Internal Rate of Return Rule 245

 A Closer Look at the Rate of Return Rule 246
 Calculating the Rate of Return for Long-Lived Projects 246
 A Word of Caution 248
 Some Pitfalls with the Internal Rate of Return Rule 248

 8.3 The Profitability Index 253
 - Capital Rationing 254 Pitfalls of the Profitability Index 254

- 8.4 The Payback Rule 255 Discounted Payback 256
- 8.5 More Mutually Exclusive Projects 256 Problem 1: The Investment Timing Decision 257 Problem 2: The Choice between Long- and Short-Lived Equipment 258

Contents

Problem 3: When to Replace an Old Machine 260

8.6 A Last Look 261

Summary 262

Questions and Problems 263

Minicase 270

Appendix: More on the IRR Rule 271

Using the IRR to Choose between Mutually Exclusive Projects 271

Using the Modified Internal Rate of Return When There Are Multiple IRRs $\ \ 271$

Chapter 9

Using Discounted Cash-Flow Analysis to Make Investment Decisions 274

- 9.1 Identifying Cash Flows 276 Discount Cash Flows, Not Profits 276 Discount Incremental Cash Flows 278 Discount Nominal Cash Flows by the Nominal Cost of Capital 281 Separate Investment and Financing Decisions 282
- 9.2 Corporate Income Taxes 283
- 9.3 An Example—Blooper Industries 283 Forecasting Blooper's Cash Flows 284

Part Three

Chapter 11

Introduction to Risk, Return, and the Opportunity Cost of Capital 332

Risk

- 11.1 Rates of Return: A Review 334
- A Century of Capital Market History 335
 Market Indexes 335
 The Historical Record 335
 Using Historical Evidence to Estimate Today's Cost of Capital 338

11.3 Measuring Risk 340

Variance and Standard Deviation 340 A Note on Calculating Variance 343 Measuring the Variation in Stock Returns 343

11.4 Risk and Diversification 345

Diversification 345 Asset versus Portfolio Risk 346 Market Risk versus Specific Risk 352 Calculating the NPV of Blooper's Mine 287 Further Notes and Wrinkles Arising from Blooper's Project 288

Summary 293 Questions and Problems 294 Minicase 302

Chapter 10

°°°

Project Analysis 304

 10.1 How Firms Organize the Investment Process to Draw on Their Competitive Strengths 306
 The Capital Budget 306
 Problems and Some Solutions 307

10.2 Reducing Forecast Bias 307

- 10.3Some "What-If" Questions308Sensitivity Analysis309Scenario Analysis312
- 10.4
 Break-Even Analysis
 312

 Accounting Break-Even Analysis
 313

 NPV Break-Even Analysis
 314

 Operating Leverage
 317
- 10.5 Real Options and the Value of Flexibility 319
 The Option to Expand 319
 A Second Real Option: The Option to Abandon 321
 A Third Real Option: The Timing Option 321
 A Fourth Real Option: Flexible Production Facilities 322

Summary 323 Questions and Problems 324 Minicase 330

 11.5 Thinking about Risk 353
 Message 1: Some Risks Look Big and Dangerous but Really Are Diversifiable 353
 Message 2: Market Risks Are Macro Risks 354
 Message 3: Risk Can Be Measured 355

Summary 356 Questions and Problems 357

Chapter 12

Risk, Return, and Capital Budgeting 362

- 12.1 Measuring Market Risk 364 Measuring Beta 364 Betas for Ford and PG&E 367 Total Risk and Market Risk 367
- 12.2 What Can You Learn from Beta? 369
 Portfolio Betas 369
 The Portfolio Beta Determines the Risk of a Diversified Portfolio 372

xxviii

12.3 Risk and Return 373
Why the CAPM Makes Sense 375
The Security Market Line 376
Using the CAPM to Estimate Expected Returns 377
How Well Does the CAPM Work? 377

Contents

 12.4 The CAPM and the Opportunity Cost of Capital 380

 The Company Cost of Capital 382

 What Determines Project Risk? 382

 Don't Add Fudge Factors to Discount Rates 383

Summary 383 Questions and Problems 384

Chapter 13 The Weighted-Average Cost of Capital and Company Valuation 392

- 13.1 Geothermal's Cost of Capital 394
- 13.2 The Weighted-Average Cost of Capital 395

Calculating Company Cost of Capital as a Weighted Average 396

Use Market Weights, Not Book Weights 398

Taxes and the Weighted-Average Cost of Capital 398 What If There Are Three (or More) Sources of Financing? 400 The NPV of Geothermal's Expansion 400 Checking Our Logic 401

- 13.3 Interpreting the Weighted-Average Cost of Capital 402
 When You Can and Can't Use WACC 402
 Some Common Mistakes 402
 How Changing Capital Structure Affects Expected Returns 403
 What Happens When the Corporate Tax Rate Is Not Zero 403
 13.4 Practical Problems: Measuring Capital Structure 403
 13.5 More Practical Problems: Estimating
 - Expected Returns
 405

 The Expected Return on Bonds
 405

 The Expected Return on Common Stock
 406

 The Expected Return on Preferred Stock
 407

 Adding It All Up
 408

 Real-Company WACCs
 408

13.6 Valuing Entire Businesses 409 Calculating the Value of the Deconstruction Business **410**

Summary 411 Questions and Problems 412 Minicase 417

Part Four Financing

Chapter 14 Introduction to Corporate Financing 420

- 14.1 Creating Value with Financing Decisions 422
- 14.2
 Patterns of Corporate Financing
 422

 Are Firms Issuing Too Much Debt?
 424
- 14.3Common Stock425Ownership of the Corporation427Voting Procedures428Classes of Stock429
- 14.4 Preferred Stock 429
- 14.5Corporate Debt430Debt Comes in Many Forms431Innovation in the Debt Market433
- 14.6 Convertible Securities 435

Summary 436 Questions and Problems 437

Chapter 15

How Corporations Raise Venture Capital and Issue Securities 440

- 15.1 Venture Capital 442 Venture Capital Companies 443
- **15.2** The Initial Public Offering445Arranging a Public Issue446Other New-Issue Procedures450The Underwriters450
- 15.3
 General Cash Offers by Public Companies
 451

 General Cash Offers and Shelf Registration
 452

 Costs of the General Cash Offer
 452

 Market Reaction to Stock Issues
 453
- 15.4 The Private Placement 454

Summary 454 Questions and Problems 455 Minicase 460 Appendix: Hotch Pot's New-Issue Prospectus 461



xxix

Debt and Payout Policy Part Five

Chapter 16

Debt Policy 466

- 16.1 How Borrowing Affects Value in a Tax-Free Economy 468 MM's Argument—A Simple Example 469 How Borrowing Affects Earnings per Share 470 How Borrowing Affects Risk and Return 472
- 16.2 Debt and the Cost of Equity 474 No Magic in Financial Leverage 476
- 16.3 Debt, Taxes, and the Weighted-Average Cost of Capital 478 Debt and Taxes at River Cruises 478 How Interest Tax Shields Contribute to the Value of Stockholders' Equity 480 Corporate Taxes and the Weighted-Average Cost of Capital 480 The Implications of Corporate Taxes for Capital Structure 482 16.4 Costs of Financial Distress 482
 - Bankruptcy Costs 483 Costs of Bankruptcy Vary with Type of Asset 484 Financial Distress without Bankruptcy 485
- 16.5 Explaining Financing Choices 487 The Trade-Off Theory 487 A Pecking Order Theory 488 The Two Faces of Financial Slack 489 Is There a Theory of Optimal Capital Structure? 490

Summary 491

Questions and Problems 492

Part Six Financial Analysis and Planning

Chapter 18

Long-Term Financial Planning 528

- 18.1 What Is Financial Planning? 530 Why Build Financial Plans? 530
- 18.2 Financial Planning Models 531 Components of a Financial Planning Model 531
- 18.3 A Long-Term Financial Planning Model for Dynamic Mattress 532 Pitfalls in Model Design 537 Choosing a Plan 538
- 18.4 External Financing and Growth 539

Summary 542 Questions and Problems 543 Minicase 549

Minicase 499 Appendix: Bankruptcy Procedures 501

Chapter 17 Payout Policy 504

- 17.1 How Corporations Pay Out Cash to Shareholders 506 How Firms Pay Dividends 507 Limitations on Dividends 507 Stock Dividends and Stock Splits 508 Stock Repurchases 509
- 17.2 The Information Content of Dividends and Repurchases 509
- 17.3 Dividends or Repurchases? The Payout Controversy 511 Dividends or Repurchases? An Example 511 Repurchases and the Dividend Discount Model 513 Dividends and Share Issues 514
- 17.4 Why Dividends May Increase Value 515
- 17.5 Why Dividends May Reduce Value 517 Taxation of Dividends and Capital Gains under Current Tax Law 518 Taxes and Payout—A Summary 518
- 17.6 Payout Policy and the Life Cycle of the Firm 518 Summary 519

Questions and Problems 520 Minicase 526

Chapter 19

Short-Term Financial Planning 550

- 19.1 Links between Long-Term and Short-Term Financing 552 Tax Strategies 553 Reasons to Hold Cash 554
- 19.2 Tracing Changes in Cash 554
- 19.3 Cash Budgeting 556 Preparing the Cash Budget 556
- 19.4 Dynamic's Short-Term Financial Plan 559 Dynamic Mattress's Financing Plan 560 Evaluating the Plan 561 A Note on Short-Term Financial Planning Models 562

Summary 564 Questions and Problems 564 Minicase 569



Chapter 20

Working Capital Management 572

- 20.1 Working Capital 574 Components of Working Capital 574 Working Capital and the Cash Cycle 574
- 20.2 Accounts Receivable and Credit Policy 577 Terms of Sale 578 Credit Agreements 579 Credit Analysis 580 The Credit Decision 581 Collection Policy 586
- 20.3 Inventory Management 587
- 20.4 Cash Management 590 Check Handling and Float 591

Other Payment Systems 592 Electronic Funds Transfer 593 International Cash Management 594

- 20.5 Investing Idle Cash: The Money Market 595 Money Market Investments 595 Calculating the Yield on Money Market Investments 596 Yields on Money Market Investments 597 The International Money Market 597
- 20.6 Managing Current Liabilities: Short-Term Debt 598 Bank Loans 598 Commercial Paper 599

Summary 600 Questions and Problems 602 Minicase 610

Part Seven Special Topics

Chapter 21 Mergers Acquisitions and Corporate

Mergers, Acquisitions, and Corporate Control 612

- 21.1 Sensible Motives for Mergers 614 Economies of Scale 616 Economies of Vertical Integration 616 Combining Complementary Resources 617 Mergers as a Use for Surplus Funds 618 Eliminating Inefficiencies 618 Industry Consolidation 618 Taxes and Cross-Border Mergers 619
- 21.2 Dubious Reasons for Mergers 619 Diversification 619 The Bootstrap Game 619
- 21.3 The Mechanics of a Merger 621 The Form of Acquisition 621 Mergers, Antitrust Law, and Popular Opposition 621
- 21.4 Evaluating Mergers 622 Mergers Financed by Cash 622 Mergers Financed by Stock 624 A Warning 625 Another Warning 625
- 21.5 The Market for Corporate Control 626
- 21.6 Method 1: Proxy Contests 626
- 21.7 Method 2: Takeovers 627
- 21.8 Method 3: Leveraged Buyouts 629 Barbarians at the Gate? 630
- 21.9 Method 4: Divestitures, Spin-Offs, and Carve-Outs 632

21.10 The Benefits and Costs of Mergers 633 Merger Waves 633

Summary 634 Questions and Problems 635 Minicase 639

Chapter 22 International Financial Management 640

- 22.1 Foreign Exchange Markets 642 Spot Exchange Rates 642 Forward Exchange Rates 644
- 22.2 Some Basic Relationships 645 Exchange Rates and Inflation 645 Real and Nominal Exchange Rates 648 Inflation and Interest Rates 648 The Forward Exchange Rate and the Expected Spot Rate 650 Interest Rates and Exchange Rates 651
- 22.3 Hedging Currency Risk 652 Transaction Risk 652 Economic Risk 653

22.4 International Capital Budgeting 654 Net Present Values for Foreign Investments 654 Political Risk 656 The Cost of Capital for Foreign Investment 657 Avoiding Fudge Factors 658

Summary 658 Questions and Problems 659 Minicase 664

23.1 Calls and Puts 668 Selling Calls and Puts 670 Payoff Diagrams Are Not Profit Diagrams 671 Financial Alchemy with Options 672 Some More Option Magic 673

Contents

- 23.2 What Determines Option Values? 674 Upper and Lower Limits on Option Values 674 The Determinants of Option Value 675 Option-Valuation Models 677
- 23.3 Spotting the Option 680 Options on Real Assets 680 Options on Financial Assets 681

Summary 684 Questions and Problems 685

Chapter 24

Risk Management 694

- 24.1 Why Hedge? 696 The Evidence on Risk Management 697
 24.2 Reducing Risk with Options 698
 24.3 Futures Contracts 698 The Mechanics of Futures Trading 701 Commodity and Financial Futures 702
 24.4 Forward Contracts 703
 24.5 Swaps 704 Interest Rate Swaps 704 Currency Swaps 706 And Some Other Swaps 707
- 24.6 Innovation in the Derivatives Market 707
- 24.7 Is "Derivative" a Four-Letter Word? 707

Summary 708 Questions and Problems 709

Part Eight Conclusion

Chapter 25 What We Do and Do Not Know about Finance 714

25.1 What We Do Know: The Six Most Important Ideas in Finance 716 Net Present Value (Chapter 5) 716 Risk and Return (Chapters 11 and 12) 716 Efficient Capital Markets (Chapter 7) 717 MM's Irrelevance Propositions (Chapters 16 and 17) 717 Option Theory (Chapter 23) 717 Agency Theory 718

25.2 What We Do Not Know: Nine Unsolved Problems in Finance 718

What Determines Project Risk and Present Value?718Risk and Return—Have We Missed Something?719

Are There Important Exceptions to the Efficient-Market Theory? 720 Is Management an Off-Balance-Sheet Liability? 720 How Can We Explain Capital Structure? 721 How Can We Resolve the Payout Controversy? 721 How Can We Explain Merger Waves? 721 What Is the Value of Liquidity? 722 Why Are Financial Systems Prone to Crisis? 722 25.3 A Final Word 723 Questions and Problems 723

Appendix A A-1

Glossary G-1

Index IND-1





CHAPTER

Goals and Governance of the Corporation

LEARNING OBJECTIVES

After studying this chapter, you should be able to:

- **1-1** Give examples of the investment and financing decisions that financial managers make.
- **1-2** Distinguish between real and financial assets.
- **1-3** Cite some of the advantages and disadvantages of organizing a business as a corporation.
- **1-4** Describe the responsibilities of the CFO, treasurer, and controller.
- **1-5** Explain why maximizing market value is the natural financial goal of the corporation.
- **1-6** Understand what is meant by "agency problems," and cite some of the ways that corporate governance helps mitigate them.
- **1-7** Understand why maximizing market value does not justify behaving unethically.

RELATED WEBSITES FOR THIS CHAPTER CAN BE FOUND IN CONNECT.



To grow from small beginnings to a major corporation, FedEx needed to make good investment and financing decisions. @Duy Phuong Nguyen/Alamy

o carry on business, a corporation needs an almost endless variety of assets. Some assets are tangible, for example, plant and machinery, office buildings, and vehicles; others are intangible, for example, brand names and patents. Corporations finance these assets by borrowing, by reinvesting profits back into the firm, and by selling additional shares to the firm's shareholders.

Financial managers, therefore, face two broad questions. First, what investments should the corporation make? Second, how should it pay for these investments? *Investment decisions* spend money. *Financing decisions* raise money for investment.

We start this chapter with examples of recent investment and financing decisions by major U.S. and foreign corporations. We review what a corporation is and describe the roles of its top financial managers. We then turn to the financial goal of the corporation, which is usually expressed as *maximizing value*, or at least adding value. Financial managers add value whenever the corporation can invest to earn a higher return than its shareholders can earn for themselves.

But managers are human beings; they cannot be perfect servants who always and everywhere maximize value. We will consider the conflicts of interest that arise in large corporations and how corporate governance helps to align the interests of managers and shareholders.

If we ask managers to maximize value, can the corporation also be a good citizen? Won't the managers be tempted to try unethical or illegal financial tricks? They sometimes may be tempted, but wise managers realize that such tricks are not just dishonest; they almost always destroy value, not increase it. More challenging for the financial manager are the gray areas where the line between ethical and unethical financial actions is hard to draw.

Finally, we look ahead to the rest of this book and look back to some entertaining snippets of financial history.

1.1 Investment and Financing Decisions

Fred Smith is best known today as the founder of FedEx. But in 1965 he was still a sophomore at Yale, where he wrote an economics term paper arguing that delivery systems were not keeping up with increasing needs for speed and dependability.¹ He later joined his stepfather at a struggling equipment and maintenance firm for air carriers. He observed firsthand the difficulties of shipping spare parts on short notice. He saw the need for an integrated air and ground delivery system with a central hub that could connect a large number of points more efficiently than a point-to-point delivery system. In 1971, at the age of 27, Smith founded Federal Express.

Like many start-up firms, Federal Express flirted again and again with failure. Smith and his family had an inheritance of a few million dollars, but this was far from enough. The young company needed to purchase and retrofit a small fleet of aging Dassault Falcon jets; build a central-hub facility; and hire and train pilots, delivery, and office staff. The initial source of capital was short-term bank loans. Because of the company's shaky financial position, the bank demanded that the planes be used as collateral and that Smith personally guarantee the loan with his own money.

In April 1973, the company went live with a fleet of 14 jets, servicing 25 U.S. cities out of its Memphis hub. By then, the company had spent \$25 million and was effectively flat broke, without enough funds to pay for its weekly delivery of jet fuel. In desperation, it managed to acquire a bank loan for \$23.7 million. This loan had to be backed by a guarantee from General Dynamics, which in return acquired an option to buy the company. (Today, General Dynamics must regret that it never exercised this option.)

In November of that year, the company finally achieved some financial stability when it raised \$24.5 million from venture capitalists, investment firms that provide funds and advice to young companies in return for a partial ownership share. Eventually, venture capitalists invested about \$90 million in Federal Express.

In 1977, private firms were allowed for the first time to compete with the Postal Service in package delivery. Federal Express responded by expanding its operations. It acquired seven Boeing 727s, each with about seven times the capacity of the Falcon jets. To pay for these new investments, Federal Express raised about \$19 million by selling shares of stock to the general public in an *initial public offering (IPO)*. The new stockholders became part-owners of the company in proportion to the number of shares they purchased.

From this point on, success followed success, and the company invested heavily to expand its air fleet as well as its supporting infrastructure. It introduced an automated shipping system and a bar-coded tracking system. In 1994, it launched its **fedex.com** website for online package tracking. It opened several new hubs across the United States as well as in Canada, France, the Philippines, and China. In 2007, FedEx (as the company was now called) became the world's largest airline measured by number of planes. FedEx also invested in other companies, capped by the acquisition of TNT Express for \$4.4 billion in 2016. By 2017, FedEx had 400,000 employees, annual revenue of \$60 billion, and a stock market value of \$67 billion. Its name had become a verb—to "FedEx a package" was to ship it overnight.

Even in retrospect, FedEx's success was hardly a sure thing. Fred Smith's idea was inspired, but its implementation was complex and difficult. FedEx had to make *good investment decisions*. In the beginning, these decisions were constrained by lack of financing. For example, used Falcon jets were the only option, given the young company's precarious financial position. At first it could service only a short list of major cities. As the company grew, its investment decisions became more complex. Which type of planes should it buy? When should it expand coverage to Europe and Asia?

¹ Legend has it that Smith received a grade of C on this paper. In fact, he doesn't remember the grade.

How many operations hubs should it build? What computer and tracking systems were necessary to keep up with the increasing package volume and geographic coverage? Which companies should it acquire as it expanded its range of services?

FedEx also needed to make *good financing decisions*. For example, how should it raise the money it needed for investment? In the beginning, these choices were limited to family money and bank loans. As the company grew, its range of choices expanded. Eventually it was able to attract funding from venture capitalists, but this posed new questions. How much cash did the firm need to raise from the venture capitalists? How big a share in the firm would the venture capitalists demand in return? The initial public offering of stock prompted similar questions. How many shares should the company try to sell? At what price? As the company grew, it raised more funds by borrowing money from its banks and by selling publicly traded bonds to investors. At each point, it needed to decide on the proper form and terms of financing as well as the amounts to be raised.

In short, FedEx needed to be *good at finance*. It had a head start over potential competitors, but a series of bad financial decisions would have sunk the company. No two companies' histories are the same, but, like FedEx, all successful companies must make good investment and financing decisions. And, as with FedEx, those decisions range from prosaic and obvious to difficult and strategically crucial.

Let's widen our discussion. Table 1.1 gives an example of a recent investment and financing decision for 10 corporations. Five are U.S. corporations and five are foreign. We have chosen very large public corporations that you are likely to be familiar with. You may have shopped at Walmart, posted a picture on Facebook, or dreamed of buying a Ferrari.

Take a look at the decisions now. We think you will agree that they appear sensible at least there is nothing obviously wrong with them. But if you are new to finance, it may be difficult to think about why these companies made these decisions and not others.

| Company | Recent Investment Decisions | Recent Financing Decisions |
|----------------------------|---|---|
| Delta Air Lines (U.S.) | Places order for 100 Airbus A321 airliners. | lssues \$1 billion 5-year bond. |
| ExxonMobil (U.S.) | Announces decision to proceed with development of a huge offshore oil discovery in Guyana. | Reinvests \$8.5 billion of the cash that it generates from operations. |
| Facebook (U.S.) | Acquires Two Big Ears, a British virtual reality audio company | Leases large new office building in San Francisco. |
| Fiat Chrysler (Italy) | Spins off its Ferrari luxury car unit. | Repays \$1.8 billion of bank debt. |
| GlaxoSmithKline (U.K.) | Spends \$3.6 billion on research and development for new drugs. | Issues additional short-term euro debt. |
| Lenovo (China) | Announces plans to build a new manufacturing facility in India to produce PCs and smartphones | Issues \$500 million of dollar bonds and \$850 million of preferred shares. |
| LVMH ² (France) | Acquires high-end perfumery, Maison Francis Kurkdjian | Partly finances acquisitions by issue of debt. |
| Procter & Gamble (U.S.) | Spends over \$7 billion on advertising. | Buys back \$4.6 billion of stock and pays a \$7.2 billion dividend. |
| Toshiba (Japan) | Agrees to sell off its bankrupt U.S. nuclear business for \$4.6 billion. | Makes \$5.4 billion issue of common stock. |
| Walmart (U.S.) | Announces plan to invest \$800 million in Chile over 3 years. | Announces offer to buy back some of its bonds. |

TABLE 1.1 Examples of recent investment and financing decisions by major public corporations

² LVMH (Moët Hennessy Louis Vuitton) markets perfumes and cosmetics, wines and spirits, leather goods, watches, and other luxury products. And, yes, we know what you are thinking, but "LVMH" really is short for "Moët Hennessy Louis Vuitton."

capital budgeting or capital expenditure (CAPEX) decision Decision to invest in tangible or intangible assets.

The Investment (Capital Budgeting) Decision

Investment decisions, such as those shown in Table 1.1, are also called **capital budgeting or capital expenditure (CAPEX) decisions.** Some of the investments in the table, such as ExxonMobil's oil platforms or Lenovo's new factory, involve tangible assets—assets that you can touch and kick. Others involve intangible assets, such as research and development (R&D), advertising, and the design of computer software. For example, major pharmaceutical manufacturers invest billions every year on R&D for new drugs.

Sometimes investments can have very-long-term consequences. For example, many U.S. nuclear power plants, which were initially licensed by the Nuclear Regulatory Commission to operate for 40 years, are now being relicensed for 20 more years, and may be able to operate efficiently for 80 years overall. Other investments may pay off in only a few months. For example, with the approach of the Christmas holidays, Walmart spends nearly \$50 billion to stock up its warehouses and retail stores. As the goods are sold over the following months, the company recovers its investment in these inventories.

The world of business can be intensely competitive, and corporations prosper only if they can keep launching new products or services. In some cases, the costs and risks of doing so are amazingly large. For example, the cost of developing the Gorgon natural gas field in Australia has been estimated at over \$40 billion. It's not surprising that this cost is being shared among several major energy companies. But do not think of companies as making billion-dollar investments on a daily basis. Most investment decisions are smaller, such as the purchase of a truck, machine tool, or computer system. Corporations make thousands of such investments each year. The cumulative amount of these small expenditures can be just as large as the occasional jumbo investments, such as those shown in Table 1.1.

Not all investments succeed. In October 2011 Hewlett-Packard (HP) paid \$11.1 billion to acquire the British software company Autonomy. Just 13 months later, HP wrote down the value of this investment by \$8.8 billion. HP claimed that it was misled by improper accounting at Autonomy. Nevertheless, the Autonomy acquisition was a disastrous investment for HP. HP's CEO was fired in short order.

There are no free guarantees in finance. But you can tilt the odds in your favor if you learn the tools of investment analysis and apply them intelligently. We cover these tools in detail later in this book.

The Financing Decision

The financial manager's second main responsibility is to raise the money that the firm requires for its investments and operations. This is the **financing decision**. When a company needs to raise money, it can invite investors to put up cash in exchange for a share of future profits, or it can promise to pay back the investors' cash plus a fixed rate of interest. In the first case, the investors receive shares of stock and become shareholders, part-owners of the corporation. The investors in this case are referred to as *equity investors*, who contribute *equity financing*. In the second case, the investors are lenders, that is, *debt investors*, who one day must be repaid. The choice between debt and equity financing is often called the *capital structure decision*. Here "capital" refers to the firm's sources of long-term financing. A firm that is seeking to raise long-term financing is said to be "raising capital."

Notice the essential difference between the investment and financing decisions. When the firm invests, it acquires **real assets**, which are then used to produce the firm's goods and services. The firm finances its investment in real assets by issuing **financial assets** to investors. A share of stock is a financial asset, which has value as a claim on the firm's real assets and on the income that those assets will produce. A bank loan is a financial asset also. It gives the bank the right to get its money back plus interest. If the firm's operations can't generate enough income to repay the bank,

financing decision

Decision on the sources and amounts of financing.

real assets

Assets used to produce goods and services.

financial assets

Financial claims to the income generated by the firm's real assets.

the bank can force the firm into bankruptcy and stake a claim on its real assets. Financial assets that can be purchased and traded by investors in public markets are called *securities*. The shares of stock issued by the public corporations in Table 1.1 are all securities. Delta's 5-year bond in Table 1.1 also is a security. But a bank loan from JPMorgan to Delta is not called a security unless the bank resells the loan to public investors.

The firm can issue an almost endless variety of financial assets. Suppose it decides to borrow. It can issue debt to investors, or it can borrow from a bank. It can borrow for 1 year or 20 years. If it borrows for 20 years, it can reserve the right to pay off the debt early. It can borrow in Paris, receiving and promising to repay euros, or it can borrow dollars in New York. (As Table 1.1 shows, GlaxoSmithKline chose to borrow euros, but it could have borrowed U.S. dollars or British pounds instead.)

In some ways, financing decisions are less important than investment decisions. Financial managers say that "value comes mainly from the investment side of the balance sheet." Also, the most successful corporations sometimes have the simplest financing strategies. Take Microsoft as an example. It is one of the world's most valuable corporations. In mid-2018, Microsoft shares traded for \$94 each. There were 7.71 billion shares outstanding. Therefore Microsoft's market value—its *market capitalization* or *market cap*—was $7.71 \times $94 = 725 billion. Where did this market value come from? It came from Microsoft's products, from its brand name and worldwide customer base, from its R&D, and from its ability to make profitable future investments. It did not come from sophisticated financing. Microsoft's financing strategy is very simple: It finances almost all investment by retaining and reinvesting operating cash flow.

Financing decisions may not add much value compared to good investment decisions, but they can destroy value if they are stupid or ambushed by bad news. For example, when a consortium of investment companies bought the energy giant TXU in 2007, the company took on an additional \$40 billion in debt. This may not have been a stupid decision, but it did prove fatal. The consortium did not foresee the expansion of shale gas production and the resulting sharp fall in natural gas and electricity prices. By April 2014 the company (renamed Energy Future Holdings) was bankrupt.

1.1 Self-Test

Are the following capital budgeting or financing decisions? (*Hint:* In one case the answer is "both.")

- a. Intel decides to spend \$7 billion to develop a new microprocessor factory.
- b. BMW borrows 350 million euros (€350 million) from Deutsche Bank.
- c. Royal Dutch Shell constructs a pipeline to bring natural gas onshore from a production platform in Australia.
- d. Avon spends €200 million to launch a new range of cosmetics in European markets.
- e. Pfizer issues new shares to buy a small biotech company.

We have emphasized the financial manager's responsibility for two decisions:

The investment decision = purchase of real assets The financing decision = sale of financial assets

But this is an oversimplification because the financial manager is also involved in many other day-to-day activities that are essential to the smooth operation of a business. For example, if the firm sells goods or services on credit, it needs to make sure

that its customers pay on time. Corporations that operate internationally must constantly transfer cash from one currency to another. And the manager must keep an eye on the risks that the firm runs and ensure that they don't land the firm in a pickle.

Self-Test 1.2

Which of the following are financial assets, and which are real assets?

- a. A patent.
- b. A share of stock issued by Wells Fargo Bank.
- c. A blast furnace in a steelmaking factory.
- d. A mortgage loan taken out to help pay for a new home.
- e. After a successful advertising campaign, potential customers trust FedEx to deliver packages promptly and reliably.
- f. An IOU ("I owe you") from your brother-in-law.

What Is a Corporation?

We have been referring to "corporations." But before going too far or too fast, we need to offer some basic definitions.

A corporation is a distinct, permanent legal entity. Suppose you decide to create a new corporation.³ You would work with a lawyer to prepare *articles of incorporation*, which set out the purpose of the business and how it is to be financed, managed, and governed. These articles must conform to the laws of the state in which the business is incorporated. For many purposes, the corporation is considered a resident of its state. For example, it can enter into contracts, borrow or lend money, and sue or be sued. It pays its own taxes (but it cannot vote!).

A corporation's owners are called *shareholders* or *stockholders*.⁴ The shareholders do not directly own the business's real assets (factories, oil wells, stores, etc.). Instead they have indirect ownership via financial assets (the shares of the corporation).

A corporation is legally distinct from the shareholders. Therefore, the shareholders have **limited liability** and cannot be held personally responsible for the corporation's debts. When the U.S. financial corporation Lehman Brothers failed in 2008, no one demanded that its stockholders put up more money to cover Lehman's massive debts. Shareholders can lose their entire investment in a corporation, but no more.

limited liability

Example

corporation

A business organized as a

separate legal entity

owned by stockholders.

The owners of a corporation are not personally liable for its obligations.

1.1 ▶

Business Organization

Suppose you buy a building and open a restaurant. You have invested in the building itself, kitchen equipment, dining-room furnishings, plus various other assets. If you do not incorporate, you own these assets personally, as the sole proprietor of the business. If you have borrowed money from a bank to start the business, then you are personally responsible for this debt. If the business loses money and cannot pay the bank, then the bank can demand

³ In the United States, corporations are identified by the label "Corporation," "Incorporated," or "Inc.," as in Caterpillar Inc. The United Kingdom identifies public corporations by "plc" (short for "Public Limited Corporation"). French corporations have the suffix "SA" ("Société Anonyme"). The corresponding labels in Germany are "GmbH" ("Gesellschaft mit beschränkter Haftung") and "AG" ("Aktiengesellschaft").

⁴ "Shareholder" and "stockholder" mean exactly the same thing and are used interchangeably.

that you raise cash by selling other assets—your car or house, for example—in order to repay the loan. But if you incorporate the restaurant business, and then the *corporation* borrows from the bank, your other assets are shielded from the restaurant's debts. Of course, incorporation also means that the bank will be more cautious in lending to you because it will have no recourse to your other assets.⁵

Notice that if you incorporate your business, you exchange direct ownership of its real assets (the building, kitchen equipment, etc.) for indirect ownership via financial assets (the shares of the new corporation).

When a corporation is first established, its shares may be privately owned by a small group of investors, perhaps the company's managers and a few backers. In this case, the shares are not publicly traded and the company is said to be *closely held*. Eventually, when the firm grows and new shares are issued to raise additional capital, its shares are traded in public markets such as the New York Stock Exchange. Such corporations are known as *public companies*. Most well-known corporations in the United States are public companies with widely dispersed shareholdings. In other countries, it is more common for large corporations to remain in private hands, and many public companies may be controlled by just a handful of investors.

A large public corporation may have hundreds of thousands of shareholders, who together own the business. An individual may have 100 shares, receive 100 votes, and be entitled to a tiny fraction of the firm's income and value. On the other hand, a pension fund or insurance company may own millions of shares, receive millions of votes, and have a correspondingly large stake in the firm's performance.

Public shareholders cannot possibly manage or control the corporation directly. Instead, they elect a *board of directors*, who in turn appoint the top managers and monitor their performance. This *separation of ownership and control* gives corporations permanence. Even if managers quit or are dismissed, the corporation survives. Today's stockholders can sell all their shares to new investors without disrupting the operations of the business. Corporations can, in principle, live forever, and in practice they may survive many human lifetimes. One of the oldest corporations is the Hudson's Bay Company, which was formed in 1670 to profit from the fur trade between northern Canada and England. The company still operates as one of Canada's leading retail chains.

The separation of corporate ownership and control can also have a downside, for it can open the door for managers and directors to act in their own interests rather than in the stockholders' interest. We return to this problem later in the chapter.

There are other disadvantages to being a corporation. One is the cost, in both time and money, of managing the corporation's legal machinery. These costs are particularly burdensome for small businesses.

There is also an important tax drawback to corporations in the United States. Because the corporation is a separate legal entity, it is taxed separately. So corporations pay tax on their profits, and shareholders are taxed again when they receive dividends from the company or sell their shares at a profit. By contrast, income generated by businesses that are not incorporated is taxed just once as personal income.⁶

Other Forms of Business Organization

Corporations do not have to be prominent, multinational businesses such as those listed in Table 1.1. You can organize a local plumbing contractor or barber shop as a



mhhe.com/brealey10e

⁵ The bank may ask you to put up personal assets as collateral for the loan to your restaurant corporation. But it has to ask and get your agreement. It doesn't have to ask if your business is a sole proprietorship.

⁶ The U.S. tax system is somewhat unusual in this respect. To avoid taxing the same income twice, many other countries give shareholders at least some credit for the taxes that the corporation has already paid.